Author's response to reviews

Title: Satisfaction with primary care provider choice and associated trust

Authors:

Ming Ying L Chu-Weininger (chuweininger@hotmail.com)
Rajesh Balkrishnan (balkrishnan.1@osu.edu)

Version: 5 Date: 19 October 2006

Dear BioMed Central Editorial Production Team,

Attached are the revised manuscript with 3 Tables, and a Figure on a separate document, following the formatting changes requested:
1) Authors qualifications removed from manuscript and submission system.
2) Discussions section changed to "Discussions and Conclusions"
3) All author names listed in the references
4) Tables submitted within the manuscript rather than separate files
5) Figure title no longer numbered
6) Figure cropped to minimize white space around images.

Thanks you for your assistance throughout the manuscript submission process.

Sincerely,
Lisa Chu-Weininger